Author's response to reviews

Title: Increasing children’s physical activity through a teaching-assistant led extracurricular intervention: Process evaluation of the Action 3:30 randomised feasibility trial

Authors:

Russell Jago (Russ.Jago@bristol.ac.uk)
Simon J Sebire (Simon.Sebire@bristol.ac.uk)
Ben Davies (Ben.Davies@bristol.ac.uk)
Lesley Wood (Lesley.Wood@bristol.ac.uk)
Kathryn Banfield (Kate.Banfield@bristol.ac.uk)
Mark J Edwards (M.J.Edwards@bristol.ac.uk)
Jane E Powell (jane.powell@uwe.ac.uk)
Alan A Montgomery (Alan.Montgomery@nottingham.ac.uk)
Janice L Thompson (j.thompson.1@bham.ac.uk)
Kenneth R Fox (K.R.Fox@bristol.ac.uk)

Version: 3
Date: 1 September 2014

Author's response to reviews:

We have now highlighted on lines 106-108 and 140-141 that the children were in school years 5 and 6 and so were between 9 and 11 years of age.